Author’s response to reviews

Title: Association between self-perception of aging, view of cancer and health of older patients in oncology: A one-year longitudinal study

Authors:

Sarah Schroyen (sarah.schroyen@ulg.ac.be)

Pierre Missotten (pmissotten@ulg.ac.be)

Guy Jérusalem (g.jerusalem@chu.ulg.ac.be)

Marjan Van den Akker (marjan.vandenakker@med.kuleuven.be)

Frank Buntinx (frank.buntinx@med.kuleuven.be)

Stéphane Adam (stephane.adam@ulg.ac.be)

Version: 1 Date: 26 Oct 2016

Author’s response to reviews:

Editor's comments:

1- Please indicate whether written informed consent was obtained from the human subjects. If written human subject consent were not necessary, please state this in the manuscript and include a brief explanation.

We added the following sentence in the methodology : "Written informed consent was obtained from the patients". We also added this information under the section : "Ethics approval and consent to participate".